\begin{table}%t1
\par
\caption{\label{embtab:sourcelist}Source list. }
 \scriptsize%\centering
\par
 \begin{tabular}{rlccccc}
\hline\hline\noalign{\smallskip}
\multicolumn{7}{c}{{Embedded sources}} \\
 {\#} & Source\hfill
                    & RA
                    & Dec
                    & AOR
                    & Tint [s]
                    & $D$\tablefootmark{a}
                    \\
 & 
               & 
               & 
               & 
               & (SL/SH/LH)*ndith.
               & [pc]
               \\
\hline
\T       1 & LDN1448\ IRS1 & \hfill\RA{3}{25}{09}{4} & \hfill\Dec{30}{46}{21}{7}\hfill & 5656832\hfill & (14*1/31*2/60*1)*2 & 250 \\
       2 & LDN1448\ NA & \hfill\RA{3}{25}{36}{5} & \hfill\Dec{30}{45}{21}{2}\hfill & 5828096\hfill & (60*1/121*2/60*1)*2 & 250 \\
       3 & IRAS\ 03245+3002 & \hfill\RA{3}{27}{39}{0} & \hfill\Dec{30}{12}{59}{4}\hfill & 6368000\hfill & (14*4/31*4/60*1)*2 & 250 \\
       4 & L1455\ SMM1 & \hfill\RA{3}{27}{43}{2} & \hfill\Dec{30}{12}{28}{8}\hfill & 15917056\hfill & (60*2/121*2/--)*2 & 250 \\
       5 & L1455\ IRS3 & \hfill\RA{3}{28}{00}{4} & \hfill\Dec{30}{08}{01}{3}\hfill & 15917568\hfill & (14*4/121*2/--)*2 & 250 \\
       6 & IRAS\ 03254+3050 & \hfill\RA{3}{28}{34}{5} & \hfill\Dec{31}{00}{51}{1}\hfill & 11827200\hfill & (14*2/31*2/60*1)*2 & 250 \\
       7 & IRAS\ 03271+3013 & \hfill\RA{3}{30}{15}{2} & \hfill\Dec{30}{23}{48}{8}\hfill & 5634304\hfill & (14*2/31*4/60*1)*2 & 250 \\
       8 & IRAS\ 03301+3111 & \hfill\RA{3}{33}{12}{8} & \hfill\Dec{31}{21}{24}{1}\hfill & 5634560\hfill & (14*1/6*2/14*2)*2 & 250 \\
       9 & B1-a & \hfill\RA{3}{33}{16}{7} & \hfill\Dec{31}{07}{55}{2}\hfill & 15918080\hfill & (14*4/31*2/--)*2 & 250 \\
      10 & B1-c & \hfill\RA{3}{33}{17}{9} & \hfill\Dec{31}{09}{31}{0}\hfill & 13460480\hfill & (14*2/--/--)*2 & 250 \\
      11 & SSTc2d\ J033327.3+310710 & \hfill\RA{3}{33}{27}{3} & \hfill\Dec{31}{07}{10}{2}\hfill & 15918336\hfill & (60*3/121*2/--)*2 & 250 \\
      12 & HH\ 211-mm & \hfill\RA{3}{43}{56}{8} & \hfill\Dec{32}{00}{50}{4}\hfill & 5826304\hfill & (--/31*2/60*2)*2 & 250 \\
      13 & IRAS\ 03439+3233 & \hfill\RA{3}{47}{05}{4} & \hfill\Dec{32}{43}{08}{4}\hfill & 5635072\hfill & (14*1/121*2/60*1)*2 & 250 \\
      14 & IRAS\ 03445+3242 & \hfill\RA{3}{47}{41}{6} & \hfill\Dec{32}{51}{43}{9}\hfill & 5635328\hfill & (14*1/6*2/14*1)*2 & 250 \\
      15 & IRAS\ 08242-5050 & \hfill\RA{8}{25}{43}{8} & \hfill\Dec{-51}{00}{35}{6}\hfill & 5638912\hfill & (14*1/6*2/14*1)*2 & 460 \\
      16 & IRAS\ 08261-5100 & \hfill\RA{8}{27}{38}{9} & \hfill\Dec{-51}{10}{37}{2}\hfill & 5638912\hfill & (14*1/6*2/14*1)*2 & 400 \\
      17 & Ced\ 110\ IRS4 & \hfill\RA{11}{06}{46}{6} & \hfill\Dec{-77}{22}{32}{5}\hfill & 5639680/12692224\hfill & (--/121*2/60*2)*2 & 178 \\
      18 & Ced\ 110\ IRS6 & \hfill\RA{11}{07}{09}{2} & \hfill\Dec{-77}{23}{04}{2}\hfill & 5639680/12696832\hfill & (--/121*2/60*2)*2 & 178 \\
      19 & IRAS\ 12553-7651 & \hfill\RA{12}{59}{06}{6} & \hfill\Dec{-77}{07}{40}{1}\hfill & 9830912\hfill & (14*1/31*1/60*1)*2 & 178 \\
      20 & IRAS\ 15398-3359 & \hfill\RA{15}{43}{02}{3} & \hfill\Dec{-34}{09}{06}{8}\hfill & 5828864\hfill & (14*2/31*6/60*1)*2 & 100 \\
      21 & GSS30-IRS1 & \hfill\RA{16}{26}{21}{4} & \hfill\Dec{-24}{23}{04}{2}\hfill & 5647616/12699392\hfill & (--/31*1/14*2)*2 & 125 \\
      22 & GY23 & \hfill\RA{16}{26}{24}{1} & \hfill\Dec{-24}{24}{48}{2}\hfill & 5647616/12699136\hfill & (--/31*1/14*2)*2 & 125 \\
      23 & VLA\ 1623-243 & \hfill\RA{16}{26}{26}{4} & \hfill\Dec{-24}{24}{30}{2}\hfill & 9828096/12695040\hfill & (--/121*4/60*4)*2 & 125 \\
      24 & WL12 & \hfill\RA{16}{26}{44}{2} & \hfill\Dec{-24}{34}{48}{4}\hfill & 5647616/12699136\hfill & (--/31*1/14*2)*2 & 125 \\
      25 & IRS37 & \hfill\RA{16}{27}{17}{6} & \hfill\Dec{-24}{28}{56}{6}\hfill & 5647616/12698880\hfill & (--/31*1/14*2)*2 & 125 \\
      26 & WL6 & \hfill\RA{16}{27}{21}{8} & \hfill\Dec{-24}{29}{53}{2}\hfill & 5647616/12698880\hfill & (--/31*1/14*2)*2 & 125 \\
      27 & Elias32 & \hfill\RA{16}{27}{28}{4} & \hfill\Dec{-24}{27}{21}{2}\hfill & 12664320/12698624\hfill & (--/31*2/60*2)*2 & 125 \\
      28 & VSSG17 & \hfill\RA{16}{27}{30}{2} & \hfill\Dec{-24}{27}{43}{6}\hfill & 5647616/12698624\hfill & (--/31*1/14*2)*2 & 125 \\
      29 & IRS63 & \hfill\RA{16}{31}{35}{7} & \hfill\Dec{-24}{01}{29}{6}\hfill & 9827840/12676608\hfill & (--/31*1/14*1)*2 & 125 \\
      30 & L1689-IRS5 & \hfill\RA{16}{31}{52}{1} & \hfill\Dec{-24}{56}{15}{4}\hfill & 12664064/12704256\hfill & (--/31*1/60*1)*2 & 125 \\
      31 & IRAS\ 16293-2422B & \hfill\RA{16}{32}{22}{6} & \hfill\Dec{-24}{28}{32}{2}\hfill & 15735808\hfill & (60*1/121*4/60*4)*2 & 125 \\
      32 & IRAS\ 16293-2422 & \hfill\RA{16}{32}{22}{9} & \hfill\Dec{-24}{28}{36}{1}\hfill & 11826944\hfill & (60*1/121*4/60*4)*2 & 125 \\
      33 & RNO\ 91 & \hfill\RA{16}{34}{29}{3} & \hfill\Dec{-15}{47}{01}{3}\hfill & 5650432\hfill & (14*1/31*1/14*2)*2 & 140 \\
      34 & SSTc2d\ J182916.2+01822 & \hfill\RA{18}{29}{16}{2} & \hfill\Dec{0}{18}{22}{7}\hfill & 13210112\hfill & (14*1/31*2/--)*2 & 260 \\
      35 & Serp-S68N & \hfill\RA{18}{29}{48}{1} & \hfill\Dec{1}{16}{42}{6}\hfill & 9828608\hfill & (--/121*4/60*4)*2 & 260 \\
      36 & Serp-SMM4 & \hfill\RA{18}{29}{56}{6} & \hfill\Dec{1}{13}{15}{2}\hfill & 9828608\hfill & (--/121*4/60*4)*2 & 260 \\
      37 & EC88 & \hfill\RA{18}{29}{57}{6} & \hfill\Dec{1}{13}{00}{5}\hfill & 9407232\hfill & (14*1/121*1/60*2)*2 & 260 \\
      38 & Serp-SMM3 & \hfill\RA{18}{29}{59}{2} & \hfill\Dec{1}{14}{00}{2}\hfill & 9828608\hfill & (--/121*4/60*4)*2 & 260 \\
      39 & R\ CrA\ IRS5 & \hfill\RA{19}{01}{48}{0} & \hfill\Dec{-36}{57}{21}{6}\hfill & 9835264\hfill & (14*1/6*2/--)*2 & 130 \\
      40 & CrA\ IRS7\ A & \hfill\RA{19}{01}{55}{3} & \hfill\Dec{-36}{57}{22}{0}\hfill & 9835008\hfill & (14*1/31*3/60*2)*2 & 170 \\
      41 & CrA\ IRS7\ B & \hfill\RA{19}{01}{56}{4} & \hfill\Dec{-36}{57}{28}{1}\hfill & 9835008\hfill & (14*1/31*3/60*2)*2 & 170 \\
      42 & CrA\ IRAS32 & \hfill\RA{19}{02}{58}{7} & \hfill\Dec{-37}{07}{34}{7}\hfill & 9832192\hfill & (60*1/31*8/--)*2 & 170 \\
      43 & IRAS\ 23238+7401 & \hfill\RA{23}{25}{46}{7} & \hfill\Dec{74}{17}{37}{3}\hfill & 9833728\hfill & (14*1/31*8/60*2)*2 & 250 \\
 \hline
\multicolumn{7}{c}{{Unknown sources}} \\
 {\#} & Source\hfill
                    & RA
                    & Dec
                    & AOR
                    & Tint [s]
                    & $D$\tablefootmark{a}
                    \\
 & 
               & 
               & 
               & 
               & (SL/SH/LH)*ndith.
               & [pc]
               \\
\hline
\T          44 & IRAS\ 13546-3941 & \hfill\RA{13}{57}{38}{9} & \hfill\Dec{-39}{56}{00}{2}\hfill & 5642752\hfill & (14*1/31*2/60*1)*2 & 630 \\
          45 & WL19 & \hfill\RA{16}{27}{11}{7} & \hfill\Dec{-24}{38}{32}{3}\hfill & 9829888/12692992\hfill & (--/31*2/14*2)*2 & 125 \\
          46 & IRAS\ 16544-1604 & \hfill\RA{16}{57}{19}{7} & \hfill\Dec{-16}{09}{23}{8}\hfill & 5826816\hfill & (--/--/14*4)*2 & 140 \\
          47 & SSTc2d\ J182813.2+00313 & \hfill\RA{18}{28}{13}{2} & \hfill\Dec{0}{03}{13}{0}\hfill & 13210368\hfill & (14*1/31*2/--)*2 & 260 \\
          48 & SSTc2d\ J182849.4+00604 & \hfill\RA{18}{28}{49}{4} & \hfill\Dec{-0}{06}{04}{7}\hfill & 13210624\hfill & (14*1/31*1/--)*2 & 260 \\
          49 & SSTc2d\ J182901.8+02954 & \hfill\RA{18}{29}{01}{8} & \hfill\Dec{0}{29}{54}{2}\hfill & 13210112\hfill & (14*1/31*2/--)*2 & 260 \\
          50 & SSTc2d\ J182914.8+00424 & \hfill\RA{18}{29}{14}{8} & \hfill\Dec{-0}{04}{23}{9}\hfill & 13210112\hfill & (14*1/31*2/--)*2 & 260 \\
          51 & EC69 & \hfill\RA{18}{29}{54}{4} & \hfill\Dec{1}{15}{01}{8}\hfill & 9407232\hfill & (14*1/121*1/60*2)*2 & 260 \\
 \hline
\multicolumn{7}{c}{{Disk sources}} \\
 {\#} & Source\hfill
                    & RA
                    & Dec
                    & AOR
                    & Tint [s]
                    & $D$\tablefootmark{a}
                    \\
 & 
               & 
               & 
               & 
               & (SL/SH/LH)*ndith.
               & [pc]
               \\
\hline
\T          52 & VSSG1 & \hfill\RA{16}{26}{18}{9} & \hfill\Dec{-24}{28}{19}{6}\hfill & 5647616/12699392\hfill & (--/31*1/14*2)*2 & 125 \\
          53 & IRS14 & \hfill\RA{16}{26}{31}{0} & \hfill\Dec{-24}{31}{05}{2}\hfill & 12664576\hfill & (60*2/31*1/--)*2 & 125 \\
          54 & OphE-MM3 & \hfill\RA{16}{27}{05}{9} & \hfill\Dec{-24}{37}{08}{0}\hfill & 6370816\hfill & (60*5/31*8/60*2)*2 & 125 \\
          55 & GY224 & \hfill\RA{16}{27}{11}{2} & \hfill\Dec{-24}{40}{46}{6}\hfill & 9829888/12698880\hfill & (--/31*2/14*2)*2 & 125 \\
          56 & WL20S & \hfill\RA{16}{27}{15}{6} & \hfill\Dec{-24}{38}{45}{6}\hfill & 9829888\hfill & (--/31*2/14*2)*2 & 125 \\
          57 & IRS42 & \hfill\RA{16}{27}{21}{5} & \hfill\Dec{-24}{41}{43}{1}\hfill & 5647616/12699648\hfill & (--/31*1/14*2)*2 & 125 \\
          58 & CRBR\ 2422.8-3423 & \hfill\RA{16}{27}{24}{6} & \hfill\Dec{-24}{41}{03}{1}\hfill & 9346048\hfill & (14*1/121*1/241*2)*2 & 125 \\
          59 & IRS46 & \hfill\RA{16}{27}{29}{4} & \hfill\Dec{-24}{39}{16}{2}\hfill & 9829888/12698624\hfill & (--/31*2/14*2)*2 & 125 \\
          60 & IRS51 & \hfill\RA{16}{27}{39}{8} & \hfill\Dec{-24}{43}{15}{2}\hfill & 9829888/12699648\hfill & (--/31*2/14*2)*2 & 125 \\
          61 & EC82 & \hfill\RA{18}{29}{56}{9} & \hfill\Dec{1}{14}{46}{7}\hfill & 9407232\hfill & (14*1/121*1/60*2)*2 & 260 \\
\hline%
\end{tabular}
\tablefoot{
\tablefoottext{a}{Assumed cloud distances;
                  Chamaeleon \citep[178~pc][]{whittet97},
                  Lupus I, IV (150~pc) and Lupus III \citep[200~pc][]{comeron08},\\
                  Ophiuchus \citep[125~pc][]{degeus89},
                  Perseus \citep[250~pc, see discussion in][]{enoch06},
                  Taurus-Auriga \citep[160~pc][]{kenyon94},\\
                  Serpens \citep[260~pc][]{straizys96}.}
                 }
\end{table}